Point by point response to reviewer’s comments for Manuscript 87849:

Reviewer’s comments: The authors present a study on the correlation between CYP3A5 genotype and tacrolimus dose adjustment after kidney transplantation. This study is of scientific relevance and explores a relatively novel area of transplant medicine. The manuscript is well-written and concise. The main limitations of the study are already described by the authors, and a follow-up study of controlled nature could yield interesting results.

- Response: Thank you very much for taking the time to provide a thorough review of our manuscript. Based on the reviewer’s comments no changes were made to the manuscript.